Author’s response to reviews

Title: Parathyroid cysts: experience of a rare phenomenon at a single institution

Authors:

Peipei Xu (xpp1590304@163.com)
Xiaotian Xia (18930172917@163.com)
Meifang Li (15821955054@126.com)
Minggao Guo (guominggao203@163.com)
Zhili Yang (yangzhili2011@live.com)

Version: 3 Date: 05 Jan 2018

Author’s response to reviews:

To editor:

1. about Availability of data and materials

The datasets generated and analysed during the current study are available from the corresponding author on reasonable request.

2. about Ethics approval and consent to participate

The need for ethics approval was waived by the ethics committee of Shanghai Jiao Tong University Affiliated Sixth People’s Hospital due to the retrospective nature of the study.

3. Any tracked changes or highlighting has been removed.

Sincerely thank for spending your time on my paper.